Author's response to reviews

Title: Neurosyphilis with dementia and bilateral hippocampal atrophy on brain magnetic resonance imaging

Authors:

- Shima Mehrabian (shima_meh@yahoo.com)
- Margarita Raycheva (margiraycheva@yahoo.com)
- Martina Traykova (martina_t9@hotmail.com)
- Tonya Stankova (tonya_stankova@yahoo.com)
- Latchezar Penev (penevl@gmail.com)
- Olga Grigorova (olgag@abv.bg)
- Latchezar Traykov (traykov_l@yahoo.fr)

Version: 2 Date: 29 February 2012

Author's response to reviews: see over
Dear Sir/Madam,

I would like to submit the article entitled “Neurosyphilis with dementia and bilateral hippocampal atrophy on brain magnetic resonance imaging” to the “BMC Neurology” Journal. This case reports a rare presentation of neurolues with unexpected and unusual neuroimaging. Syphilis is known as the great imitator and it will have a broader clinical impact across more than one area of medicine.

I state that all authors have contributed to the work, agree with the presented findings, and that the work has not been published before nor is being considered for publication elsewhere. The authors report no conflicts of interest.

Additionally, I update the consent section and add a section for acknowledgements and funding.

Sincerely yours,

Dr Shima Mehrabian, MD, PhD (corresponding author)
Department of Neurology
University Hospital Alexandrovska
tel.: +359 887 42 86 76
Fax: +35929503400
e-mail: shima_meh@yahoo.com